

Manuscript NO: 00091579

Madrid, May 1st 2024

Dr. Lian-Sheng Ma, Founder and Chief Executive Officer
Baishideng Publishing Group, Inc.

Dear Editor:

Thank you for your preliminary decision regarding our invited paper entitled "Hyoid metastasis an unusual location from lung cancer.", which was sent to the World Journal of Clinical Oncology for publication as a Editorial.

I would like to thank the reviewer for their earnest efforts in reviewing the manuscript. I accepted and responded step by step to all considerations by the reviewer improving the manuscript. The changes are highlighted by yellow.

Reviewer 1:

Hyoid metastasis from lung cancer is extremely rare. This editorial remarks the importance of radiotherapy in palliative care, especially in pain control, always accompanied of systemic therapy in oncological patients;

Answer:

- Many thanks for his considerable comments.
- We truly believe in the importance of systemic therapy but in some cases systemic treatment accompanied by local treatment as radiotherapy is the best option specially in palliative care, as is reported in this editorial based in a clinical case with good pain response.

I am sending the revised manuscript and hope to receive a favorable final decision.

We look forward to hearing from you at your earliest convenience.

Sincerely,

Abrahams Ocanto, MD
Department of Radiation Oncology, Hospital Universitario San Francisco de Asís, GenesisCare
Madrid, 28002 Madrid, Spain. e-mail: abrahams.ocanto@genesiscare.es